Author's response to reviews

Title: Microvessel density as a new prognostic marker after radiotherapy in rectal cancer

Authors:

Saulius Svagzdys (s.svagzdys@gmail.com)
Vaiva Lesauskaite (vaiva.lesauskaite@med.kmu.lt)
Dainius Pavalkis (dainius.pavalkis@kmuk.lt)
Irena Nedzelskiene (irenaned@med.kmu.lt)
Darius Pranys (patkmuk@kmu.lt)
Algimantas Tamelis (kp@kmuk.lt)

Version: 3 Date: 24 February 2009

Author's response to reviews: see over
Covering letter
Microvessel density as a new prognostic marker after radiotherapy in rectal cancer

Dear editorial board,
Thank you for assessing our manuscript once more. We have made corrections according to the advices presented by Medical Editor. We would like to comment our additions.

At first we are sorry about a mistake for name of Referee 3. We don’t know what happened, but our reference manager gives that variant of referent in the list during automatic bibliography generation [1].

Reference List


We correct this reference manually to:


We added section in our Discussion concerning limitation of our study sample size (page 10).

We went through the manuscript formatting checklist and revised manuscript according the manuscript formatting checklist.

We send you corrected manuscript once more.
Respectfully
Saulius Svagzdys